\begin{table}%t2
\caption{\label{table2}Wavelengths and radiative rates for $N$-electron Fe ions.
Note: $a\pm\!b\equiv a\times 10^{\pm b}$.}
%\centering
\par
\begin{tabular}{rrrllrrrllrrrll}
\hline\hline\noalign{\smallskip}
\multicolumn{1}{c}{$N$}&\multicolumn{1}{c}{$k$}&\multicolumn{1}{c}{$i$}&\multicolumn{1}{c}{$\lambda$}&
\multicolumn{1}{c}{$A_{\rm r}(k,i)$}&
\multicolumn{1}{c}{$N$}&\multicolumn{1}{c}{$k$}&\multicolumn{1}{c}{$i$}&\multicolumn{1}{c}{$\lambda$}&
\multicolumn{1}{c}{$A_{\rm r}(k,i)$}&
\multicolumn{1}{c}{$N$}&\multicolumn{1}{c}{$k$}&\multicolumn{1}{c}{$i$}&\multicolumn{1}{c}{$\lambda$}&
\multicolumn{1}{c}{$A_{\rm r}(k,i)$} \\
\multicolumn{1}{c}{}&\multicolumn{1}{c}{}&\multicolumn{1}{c}{}&\multicolumn{1}{c}{(\AA)}&\multicolumn{1}{c}{(s$^{-1}$)}&
\multicolumn{1}{c}{}&\multicolumn{1}{c}{}&\multicolumn{1}{c}{}&\multicolumn{1}{c}{(\AA)}&\multicolumn{1}{c}{(s$^{-1}$)}&
\multicolumn{1}{c}{}&\multicolumn{1}{c}{}&\multicolumn{1}{c}{}&\multicolumn{1}{c}{(\AA)}&\multicolumn{1}{c}{(s$^{-1}$)}\\
\hline
2 &  2 &  1 & 1.8682 &  2.08~$+$~08 & 4 &  8 &  4 & 166.69 &   7.44~$+$~09 & 4 & 24 &  2 & 1.8665 &   5.27~$+$~12\\
2 &  3 &  2 & 428.23 &  3.74~$+$~08 & 4 &  8 &  5 & 313.19 &   3.74~$+$~08 & 4 & 24 &  3 & 1.8676 &   3.99~$+$~13\\
2 &  4 &  1 & 1.8595 &  4.35~$+$~13 & 4 &  9 &  3 & 121.20 &   4.74~$+$~08 & 4 & 24 &  4 & 1.8708 &   9.55~$+$~13\\
2 &  4 &  2 & 400.30 &  4.12~$+$~08 & 4 &  9 &  4 & 136.53 &   4.81~$+$~09 & 4 & 24 &  5 & 1.8807 &   2.18~$+$~13\\
2 &  5 &  1 &        &  4.11~$+$~09$^{{\rm a}}$ & 4 &  9 &  5 & 221.34 &   4.47~$+$~09 & 4 & 25 &  3 & 1.8669$^{{\rm b}}$ &   1.00~$+$~14\\
2 &  6 &  1 & 1.8554 &  6.48~$+$~09 & 4 & 10 &  3 & 95.834 &   1.72~$+$~08 & 4 & 25 &  5 & 1.8800$^{{\rm b}}$ &   2.97~$+$~13\\
2 &  6 &  2 & 271.12 &  1.42~$+$~09 & 4 & 10 &  5 & 149.21 &   3.14~$+$~10 & 4 & 26 &  3 & 1.8651 &   1.96~$+$~11\\
2 &  7 &  1 & 1.8504 &  4.61~$+$~14 & 4 & 11 &  7 & 1.9174$^{{\rm b}}$ &   8.04~$+$~12 & 4 & 26 &  4 & 1.8683 &   9.67~$+$~13\\
2 &  7 &  2 & 194.28 &  3.40~$+$~08 & 4 & 12 &  1 & 1.8797 &   3.13~$+$~13 & 4 & 26 &  5 & 1.8781 &   1.65~$+$~14\\
2 &  7 &  5 & 382.76 &  4.69~$+$~08 & 4 & 12 &  6 & 1.9141 &   3.69~$+$~12 & 4 & 27 &  2 & 1.8624$^{{\rm b}}$ &   2.73~$+$~11\\
3 &  2 &  1 & 255.11 &  1.77~$+$~09 & 4 & 12 &  7 & 1.9167 &   2.05~$+$~12 & 4 & 27 &  3 & 1.8635$^{{\rm b}}$ &   8.69~$+$~10\\
3 &  3 &  1 & 192.03 &  4.30~$+$~09 & 4 & 12 &  8 & 1.9184 &   3.51~$+$~12 & 4 & 27 &  4 & 1.8667$^{{\rm b}}$ &   1.51~$+$~14\\
3 &  3 &  2 & 776.55 &  2.08~$+$~04 & 4 & 12 &  9 & 1.9233 &   3.26~$+$~10 & 4 & 27 &  5 & 1.8765$^{{\rm b}}$ &   1.17~$+$~13\\
3 &  4 &  2 & 1.8924 &  9.27~$+$~12 & 4 & 12 & 10 & 1.9314 &   8.09~$+$~09 & 4 & 28 &  7 & 1.8859$^{{\rm b}}$ &   6.20~$+$~12\\
3 &  4 &  3 & 1.8970 &  9.52~$+$~12 & 4 & 13 &  7 & 1.9130$^{{\rm b}}$ &   3.25~$+$~12 & 4 & 28 &  8 & 1.8875$^{{\rm b}}$ &   5.35~$+$~12\\
3 &  5 &  1 & 1.8747 &  4.97~$+$~12 & 4 & 13 &  8 & 1.9146$^{{\rm b}}$ &   5.71~$+$~12 & 4 & 28 &  9 & 1.8922$^{{\rm b}}$ &   2.50~$+$~11\\
3 &  6 &  1 & 1.8738 &  1.55~$+$~13 & 4 & 13 &  9 & 1.9195$^{{\rm b}}$ &   1.40~$+$~12 & 4 & 29 &  3 & 1.8618$^{{\rm b}}$ &   8.07~$+$~11\\
3 &  7 &  1 & 1.8706$^{{\rm b}}$ &  6.16~$+$~09 & 4 & 14 &  2 & 1.8867$^{{\rm b}}$ &   3.77~$+$~12 & 4 & 29 &  4 & 1.8650$^{{\rm b}}$ &   5.99~$+$~12\\
3 &  7 &  3 & 1.8890$^{{\rm b}}$ &  1.94~$+$~08 & 4 & 14 &  3 & 1.8878$^{{\rm b}}$ &   7.64~$+$~12 & 4 & 29 &  5 & 1.8748$^{{\rm b}}$ &   5.31~$+$~13\\
3 &  8 &  1 & 1.8635 &  3.01~$+$~14 & 4 & 14 &  4 & 1.8911$^{{\rm b}}$ &   2.88~$+$~11 & 4 & 30 &  2 & 1.8576 &   1.02~$+$~12\\
3 &  9 &  1 & 1.8611 &  4.71~$+$~14 & 4 & 14 &  5 & 1.9012$^{{\rm b}}$ &   1.12~$+$~09 & 4 & 30 &  3 & 1.8586 &   8.31~$+$~09\\
3 & 10 &  2 & 1.8722 &  2.17~$+$~13 & 4 & 15 &  3 & 1.8856$^{{\rm b}}$ &   8.10~$+$~09 & 4 & 30 &  4 & 1.8618 &   1.71~$+$~12\\
3 & 10 &  3 & 1.8767 &  9.12~$+$~10 & 4 & 15 &  4 & 1.8889$^{{\rm b}}$ &   5.05~$+$~12 & 4 & 30 &  5 & 1.8716 &   6.64~$+$~14\\
3 & 11 &  1 & 1.8570 &  1.86~$+$~14 & 4 & 15 &  5 & 1.8990$^{{\rm b}}$ &   1.65~$+$~09 & 4 & 31 &  3 & 1.8562 &   4.31~$+$~10\\
3 & 12 &  1 & 1.8563 &  4.19~$+$~12 & 4 & 16 &  4 & 1.8871 &   1.02~$+$~13 & 4 & 31 &  5 & 1.8692 &   2.23~$+$~14\\
3 & 13 &  2 & 1.8699 &  4.51~$+$~10 & 4 & 17 &  1 & 1.8704 &   4.37~$+$~14 & 4 & 32 &  1 & 1.8414$^{{\rm b}}$ &   1.29~$+$~11\\
3 & 13 &  3 & 1.8744 &  1.06~$+$~13 & 4 & 17 &  6 & 1.9045 &   6.04~$+$~10 & 4 & 32 &  6 & 1.8745$^{{\rm b}}$ &   2.57~$+$~14\\
3 & 14 &  3 & 1.8728 &  3.58~$+$~13 & 4 & 17 &  7 & 1.9071 &   3.47~$+$~11 & 4 & 32 &  7 & 1.8770$^{{\rm b}}$ &   2.28~$+$~12\\
3 & 15 &  2 & 1.8633 &  3.14~$+$~14 & 4 & 17 &  8 & 1.9087 &   3.95~$+$~12 & 4 & 32 &  8 & 1.8785$^{{\rm b}}$ &   4.76~$+$~13\\
3 & 15 &  3 & 1.8678 &  3.64~$+$~13 & 4 & 17 &  9 & 1.9135 &   1.14~$+$~13 & 4 & 32 &  9 & 1.8832$^{{\rm b}}$ &   1.07~$+$~12\\
3 & 16 &  2 & 1.8628 &  5.31~$+$~14 & 4 & 17 & 10 & 1.9215 &   8.82~$+$~11 & 4 & 32 & 10 & 1.8910$^{{\rm b}}$ &   2.89~$+$~10\\
3 & 16 &  3 & 1.8672 &  1.60~$+$~14 & 4 & 18 &  3 & 1.8752 &   6.01~$+$~14 & 4 & 33 &  7 & 1.8766$^{{\rm b}}$ &   2.15~$+$~14\\
3 & 17 &  3 & 1.8660 &  2.05~$+$~14 & 4 & 18 &  5 & 1.8884 &   5.64~$+$~11 & 4 & 33 &  8 & 1.8781$^{{\rm b}}$ &   2.11~$+$~10\\
3 & 18 &  2 & 1.8578 &  9.69~$+$~12 & 4 & 19 &  2 & 1.8730$^{{\rm b}}$ &   3.80~$+$~14 & 4 & 33 &  9 & 1.8828$^{{\rm b}}$ &   2.01~$+$~13\\
3 & 18 &  3 & 1.8622 &  6.07~$+$~14 & 4 & 19 &  3 & 1.8741$^{{\rm b}}$ &   5.62~$+$~13 & 4 & 34 &  8 & 1.8776$^{{\rm b}}$ &   1.65~$+$~14\\
3 & 19 &  2 & 1.8525 &  1.03~$+$~13 & 4 & 19 &  4 & 1.8773$^{{\rm b}}$ &   1.14~$+$~14 & 4 & 34 &  9 & 1.8823$^{{\rm b}}$ &   6.69~$+$~13\\
3 & 19 &  3 & 1.8569 &  2.40~$+$~14 & 4 & 19 &  5 & 1.8873$^{{\rm b}}$ &   9.97~$+$~10 & 4 & 35 &  1 & 1.8382$^{{\rm b}}$ &   1.76~$+$~10\\
4 &  3 &  1 & 263.77 &  5.01~$+$~07 & 4 & 20 &  3 & 1.8736 &   3.14~$+$~14 & 4 & 35 &  6 & 1.8711$^{{\rm b}}$ &   4.99~$+$~13\\
4 &  4 &  1 & 211.96 &  7.71~$+$~00 & 4 & 20 &  4 & 1.8769 &   1.59~$+$~13 & 4 & 35 &  7 & 1.8736$^{{\rm b}}$ &   4.54~$+$~14\\
4 &  4 &  3 & 1079.3 &  1.05~$+$~04 & 4 & 20 &  5 & 1.8868 &   5.75~$+$~12 & 4 & 35 &  8 & 1.8751$^{{\rm b}}$ &   2.41~$+$~14\\
4 &  5 &  1 & 132.91 &  1.95~$+$~10 & 4 & 21 &  4 & 1.8757 &   2.23~$+$~14 & 4 & 35 &  9 & 1.8798$^{{\rm b}}$ &   5.27~$+$~13\\
4 &  6 &  3 & 173.32 &  1.22~$+$~10 & 4 & 22 &  2 & 1.8708 &   3.07~$+$~12 & 4 & 35 & 10 & 1.8875$^{{\rm b}}$ &   7.56~$+$~10\\
4 &  6 &  5 & 490.94 &  2.26~$+$~07 & 4 & 22 &  3 & 1.8719 &   2.61~$+$~14 & 4 & 36 &  7 & 1.8712$^{{\rm b}}$ &   5.71~$+$~11\\
4 &  7 &  2 & 147.27 &  6.52~$+$~09 & 4 & 22 &  4 & 1.8752 &   7.30~$+$~13 & 4 & 36 &  8 & 1.8727$^{{\rm b}}$ &   3.60~$+$~14\\
4 &  7 &  3 & 154.30 &  4.13~$+$~09 & 4 & 22 &  5 & 1.8851 &   3.23~$+$~12 & 4 & 36 &  9 & 1.8774$^{{\rm b}}$ &   1.60~$+$~14\\
4 &  7 &  4 & 180.04 &  4.42~$+$~09 & 4 & 23 &  3 & 1.8703$^{{\rm b}}$ &   3.47~$+$~13 & 4 & 37 &  7 & 1.8697$^{{\rm b}}$ &   2.27~$+$~14\\
4 &  7 &  5 & 363.91 &  8.10~$+$~06 & 4 & 23 &  4 & 1.8735$^{{\rm b}}$ &   4.70~$+$~14 & 4 & 38 &  1 & 1.8340$^{{\rm b}}$ &   3.42~$+$~10\\
4 &  8 &  3 & 144.39 &  5.31~$+$~09 & 4 & 23 &  5 & 1.8834$^{{\rm b}}$ &   1.49~$+$~13 & 4 & 38 &  6 & 1.8667$^{{\rm b}}$ &   2.77~$+$~12\\
\hline
4 & 38 &  7 & 1.8692$^{{\rm b}}$ &  1.19~$+$~13 & 5 & 13 & 10 & 230.28 &   3.27~$+$~09$^{{\rm c}}$ & 5 & 21 & 13 & 1.9343 &   4.11~$+$~12\\
4 & 38 &  8 & 1.8708$^{{\rm b}}$ &  2.59~$+$~14 & 5 & 14 &  3 & 85.831 &   2.07~$+$~08$^{{\rm c}}$ & 5 & 21 & 14 & 1.9397 &   1.11~$+$~12\\
4 & 38 &  9 & 1.8754$^{{\rm b}}$ &  1.55~$+$~13 & 5 & 14 &  4 & 90.136 &   5.50~$+$~07$^{{\rm c}}$ & 5 & 21 & 15 & 1.9419 &   5.13~$+$~08\\
4 & 38 & 10 & 1.8831$^{{\rm b}}$ &  3.01~$+$~13 & 5 & 14 &  6 & 120.00 &   2.83~$+$~10$^{{\rm c}}$ & 5 & 22 &  2 & 1.8851 &   1.97~$+$~14\\
4 & 39 &  7 & 1.8670$^{{\rm b}}$ &  7.40~$+$~12 & 5 & 14 &  8 & 139.67 &   2.32~$+$~09$^{{\rm c}}$ & 5 & 22 & 11 & 1.9264 &   1.17~$+$~11\\
4 & 39 &  8 & 1.8686$^{{\rm b}}$ &  1.19~$+$~12 & 5 & 14 &  9 & 169.12 &   1.14~$+$~10$^{{\rm c}}$ & 5 & 22 & 12 & 1.9316 &   1.38~$+$~12\\
4 & 39 &  9 & 1.8732$^{{\rm b}}$ &  3.88~$+$~14 & 5 & 14 & 10 & 173.22 &   2.13~$+$~09$^{{\rm c}}$ & 5 & 22 & 13 & 1.9328 &   4.84~$+$~12\\
4 & 40 &  1 & 1.8278$^{{\rm b}}$ &  3.91~$+$~11 & 5 & 15 &  3 & 81.755 &   7.37~$+$~06$^{{\rm c}}$ & 5 & 22 & 15 & 1.9403 &   4.58~$+$~11\\
4 & 40 &  6 & 1.8603$^{{\rm b}}$ &  1.88~$+$~11 & 5 & 15 &  4 & 85.651 &   3.09~$+$~08$^{{\rm c}}$ & 5 & 23 &  1 & 1.8782 &   2.27~$+$~12\\
4 & 40 &  7 & 1.8627$^{{\rm b}}$ &  1.70~$+$~12 & 5 & 15 &  5 & 89.730 &   1.29~$+$~08$^{{\rm c}}$ & 5 & 23 &  2 & 1.8824 &   5.35~$+$~14\\
4 & 40 &  8 & 1.8643$^{{\rm b}}$ &  8.25~$+$~12 & 5 & 15 &  6 & 112.18 &   4.92~$+$~09$^{{\rm c}}$ & 5 & 23 & 11 & 1.9236 &   2.08~$+$~09\\
4 & 40 &  9 & 1.8689$^{{\rm b}}$ &  3.85~$+$~14 & 5 & 15 &  7 & 115.14 &   1.39~$+$~10$^{{\rm c}}$ & 5 & 23 & 12 & 1.9288 &   1.76~$+$~12\\
4 & 40 & 10 & 1.8766$^{{\rm b}}$ &  2.87~$+$~14 & 5 & 15 &  8 & 129.19 &   3.78~$+$~09$^{{\rm c}}$ & 5 & 23 & 13 & 1.9299 &   4.91~$+$~12\\
5 &  2 &  1 & 845.55 &  1.49~$+$~04$^{{\rm c}}$ & 5 & 15 &  9 & 154.00 &   1.73~$+$~09$^{{\rm c}}$ & 5 & 23 & 14 & 1.9353 &   8.90~$+$~09\\
5 &  3 &  1 & 247.19 &  8.50~$+$~07$^{{\rm c}}$ & 5 & 15 & 10 & 157.38 &   1.91~$+$~10$^{{\rm c}}$ & 5 & 23 & 15 & 1.9374 &   4.14~$+$~12\\
5 &  3 &  2 & 349.30 &  1.44~$+$~07$^{{\rm c}}$ & 5 & 16 &  1 & 1.8918$^{{\rm b}}$ &   2.97~$+$~13 & 5 & 24 &  1 & 1.8752 &   3.27~$+$~12\\
5 &  4 &  1 & 217.30 &  2.10~$+$~06$^{{\rm c}}$ & 5 & 16 &  2 & 1.8960$^{{\rm b}}$ &   9.20~$+$~11 & 5 & 24 &  2 & 1.8794 &   2.64~$+$~14\\
5 &  4 &  2 & 292.46 &  8.35~$+$~06$^{{\rm c}}$ & 5 & 16 & 11 & 1.9378$^{{\rm b}}$ &   3.17~$+$~12 & 5 & 24 & 11 & 1.9204 &   1.65~$+$~09\\
5 &  5 &  2 & 253.17 &  7.23~$+$~07$^{{\rm c}}$ & 5 & 16 & 12 & 1.9431$^{{\rm b}}$ &   2.30~$+$~10 & 5 & 24 & 12 & 1.9256 &   8.62~$+$~09\\
5 &  6 &  1 & 135.81 &  1.08~$+$~10$^{{\rm c}}$ & 5 & 16 & 14 & 1.9496$^{{\rm b}}$ &   1.16~$+$~10 & 5 & 24 & 14 & 1.9321 &   4.51~$+$~12\\
5 &  6 &  2 & 161.80 &  3.22~$+$~07$^{{\rm c}}$ & 5 & 16 & 15 & 1.9519$^{{\rm b}}$ &   4.93~$+$~09 & 5 & 24 & 15 & 1.9343 &   3.72~$+$~12\\
5 &  7 &  2 & 156.02 &  6.03~$+$~09$^{{\rm c}}$ & 5 & 17 &  1 & 1.8893$^{{\rm b}}$ &   1.25~$+$~11 & 5 & 25 &  3 & 1.8845$^{{\rm b}}$ &   2.89~$+$~14\\
5 &  8 &  1 & 117.14 &  3.84~$+$~10$^{{\rm c}}$ & 5 & 17 &  2 & 1.8935$^{{\rm b}}$ &   9.76~$+$~12 & 5 & 25 &  4 & 1.8865$^{{\rm b}}$ &   7.40~$+$~11\\
5 &  8 &  2 & 135.98 &  2.60~$+$~07$^{{\rm c}}$ & 5 & 17 & 11 & 1.9352$^{{\rm b}}$ &   3.20~$+$~12 & 5 & 25 &  5 & 1.8884$^{{\rm b}}$ &   5.35~$+$~13\\
5 &  9 &  1 & 102.21 &  2.58~$+$~09$^{{\rm c}}$ & 5 & 17 & 12 & 1.9404$^{{\rm b}}$ &   2.14~$+$~11 & 5 & 25 &  6 & 1.8964$^{{\rm b}}$ &   4.34~$+$~12\\
5 &  9 &  2 & 116.27 &  3.43~$+$~10$^{{\rm c}}$ & 5 & 17 & 13 & 1.9416$^{{\rm b}}$ &   4.51~$+$~10 & 5 & 25 &  7 & 1.8972$^{{\rm b}}$ &   7.40~$+$~11\\
5 & 10 &  1 & 100.77 &  6.27~$+$~09$^{{\rm c}}$ & 5 & 17 & 14 & 1.9470$^{{\rm b}}$ &   4.39~$+$~08 & 5 & 25 &  8 & 1.9006$^{{\rm b}}$ &   9.57~$+$~11\\
5 & 10 &  2 & 114.41 &  4.44~$+$~10$^{{\rm c}}$ & 5 & 17 & 15 & 1.9492$^{{\rm b}}$ &   2.44~$+$~09 & 5 & 25 &  9 & 1.9051$^{{\rm b}}$ &   3.85~$+$~10\\
5 & 11 &  3 & 117.49 &  1.02~$+$~10$^{{\rm c}}$ & 5 & 18 &  2 & 1.8919$^{{\rm b}}$ &   2.59~$+$~13 & 5 & 25 & 10 & 1.9056$^{{\rm b}}$ &   1.97~$+$~09\\
5 & 11 &  4 & 125.71 &  1.51~$+$~10$^{{\rm c}}$ & 5 & 18 & 11 & 1.9335$^{{\rm b}}$ &   3.84~$+$~12 & 5 & 26 &  4 & 1.8862$^{{\rm b}}$ &   2.16~$+$~14\\
5 & 11 &  5 & 134.69 &  1.96~$+$~10$^{{\rm c}}$ & 5 & 18 & 12 & 1.9387$^{{\rm b}}$ &   1.19~$+$~10 & 5 & 26 &  5 & 1.8880$^{{\rm b}}$ &   2.27~$+$~12\\
5 & 11 &  6 & 192.53 &  1.42~$+$~08$^{{\rm c}}$ & 5 & 18 & 13 & 1.9399$^{{\rm b}}$ &   3.55~$+$~11 & 5 & 26 &  6 & 1.8960$^{{\rm b}}$ &   9.39~$+$~10\\
5 & 11 &  7 & 201.41 &  1.34~$+$~07$^{{\rm c}}$ & 5 & 18 & 15 & 1.9475$^{{\rm b}}$ &   4.16~$+$~09 & 5 & 26 &  7 & 1.8969$^{{\rm b}}$ &   7.80~$+$~12\\
5 & 11 &  8 & 248.73 &  1.24~$+$~08$^{{\rm c}}$ & 5 & 19 &  4 & 1.9007$^{{\rm b}}$ &   2.96~$+$~12 & 5 & 26 & 10 & 1.9053$^{{\rm b}}$ &   3.78~$+$~11\\
5 & 11 &  9 & 360.56 &  1.21~$+$~07$^{{\rm c}}$ & 5 & 19 &  5 & 1.9026$^{{\rm b}}$ &   3.27~$+$~12 & 5 & 27 &  3 & 1.8838$^{{\rm b}}$ &   2.20~$+$~14\\
5 & 11 & 10 & 379.68 &  3.07~$+$~07$^{{\rm c}}$ & 5 & 19 &  6 & 1.9108$^{{\rm b}}$ &   2.58~$+$~09 & 5 & 27 &  4 & 1.8858$^{{\rm b}}$ &   8.52~$+$~12\\
5 & 12 &  3 & 100.85 &  4.57~$+$~07$^{{\rm c}}$ & 5 & 19 &  7 & 1.9116$^{{\rm b}}$ &   2.47~$+$~10 & 5 & 27 &  6 & 1.8957$^{{\rm b}}$ &   1.05~$+$~13\\
5 & 12 &  4 & 106.85 &  2.22~$+$~09$^{{\rm c}}$ & 5 & 19 & 10 & 1.9201$^{{\rm b}}$ &   1.65~$+$~08 & 5 & 27 &  8 & 1.8999$^{{\rm b}}$ &   3.49~$+$~11\\
5 & 12 &  5 & 113.27 &  2.25~$+$~08$^{{\rm c}}$ & 5 & 20 &  1 & 1.8825 &   5.00~$+$~14 & 5 & 27 &  9 & 1.9044$^{{\rm b}}$ &   3.42~$+$~10\\
5 & 12 &  6 & 151.56 &  7.49~$+$~09$^{{\rm c}}$ & 5 & 20 &  2 & 1.8867 &   1.19~$+$~14 & 5 & 27 & 10 & 1.9049$^{{\rm b}}$ &   6.84~$+$~10\\
5 & 12 &  7 & 157.01 &  4.88~$+$~09$^{{\rm c}}$ & 5 & 20 & 11 & 1.9280 &   3.54~$+$~11 & 5 & 28 &  5 & 1.8874$^{{\rm b}}$ &   2.10~$+$~14\\
5 & 12 &  8 & 184.35 &  6.31~$+$~09$^{{\rm c}}$ & 5 & 20 & 12 & 1.9333 &   9.90~$+$~12 & 5 & 28 &  7 & 1.8962$^{{\rm b}}$ &   1.33~$+$~13\\
5 & 12 &  9 & 239.37 &  7.93~$+$~08$^{{\rm c}}$ & 5 & 20 & 14 & 1.9398 &   7.09~$+$~11 & 5 & 29 &  3 & 1.8824$^{{\rm b}}$ &   4.58~$+$~13\\
5 & 12 & 10 & 247.65 &  5.18~$+$~07$^{{\rm c}}$ & 5 & 20 & 15 & 1.9420 &   1.06~$+$~11 & 5 & 29 &  4 & 1.8844$^{{\rm b}}$ &   4.18~$+$~14\\
5 & 13 &  4 & 103.48 &  3.84~$+$~07$^{{\rm c}}$ & 5 & 21 &  1 & 1.8824 &   2.98~$+$~14 & 5 & 29 &  5 & 1.8863$^{{\rm b}}$ &   2.61~$+$~14\\
5 & 13 &  5 & 109.49 &  1.89~$+$~09$^{{\rm c}}$ & 5 & 21 &  2 & 1.8866 &   6.07~$+$~13 & 5 & 29 &  6 & 1.8943$^{{\rm b}}$ &   2.55~$+$~12\\
5 & 13 &  6 & 144.87 &  3.36~$+$~09$^{{\rm c}}$ & 5 & 21 & 11 & 1.9280 &   2.01~$+$~11 & 5 & 29 &  7 & 1.8951$^{{\rm b}}$ &   3.50~$+$~12\\
5 & 13 &  7 & 149.84 &  1.26~$+$~10$^{{\rm c}}$ & 5 & 21 & 12 & 1.9332 &   2.05~$+$~12 & 5 & 29 &  8 & 1.8985$^{{\rm b}}$ &   7.45~$+$~11\\
\hline
5 & 29 &  9 & 1.9030$^{{\rm b}}$ &   5.51~$+$~09 & 5 & 36 &  9 & 1.8902$^{{\rm b}}$ &   1.50~$+$~13 & 5 & 44 & 11 & 1.8850$^{{\rm b}}$ &   2.38~$+$~14\\
5 & 29 & 10 & 1.9035$^{{\rm b}}$ &   2.83~$+$~10 & 5 & 36 & 10 & 1.8907$^{{\rm b}}$ &   2.50~$+$~12 & 5 & 44 & 12 & 1.8900$^{{\rm b}}$ &   6.61~$+$~12\\
5 & 30 &  3 & 1.8786$^{{\rm b}}$ &   7.49~$+$~12 & 5 & 37 &  3 & 1.8694$^{{\rm b}}$ &   1.00~$+$~12 & 5 & 44 & 13 & 1.8911$^{{\rm b}}$ &   9.94~$+$~12\\
5 & 30 &  4 & 1.8806$^{{\rm b}}$ &   2.20~$+$~14 & 5 & 37 &  4 & 1.8713$^{{\rm b}}$ &   1.55~$+$~10 & 5 & 44 & 14 & 1.8963$^{{\rm b}}$ &   3.01~$+$~11\\
5 & 30 &  6 & 1.8904$^{{\rm b}}$ &   1.83~$+$~10 & 5 & 37 &  6 & 1.8811$^{{\rm b}}$ &   3.82~$+$~14 & 5 & 44 & 15 & 1.8984$^{{\rm b}}$ &   1.33~$+$~12\\
5 & 30 &  8 & 1.8946$^{{\rm b}}$ &   1.09~$+$~11 & 5 & 37 &  8 & 1.8852$^{{\rm b}}$ &   1.44~$+$~14 & 5 & 45 &  1 & 1.8405$^{{\rm b}}$ &   3.54~$+$~09\\
5 & 30 &  9 & 1.8991$^{{\rm b}}$ &   2.74~$+$~12 & 5 & 37 &  9 & 1.8897$^{{\rm b}}$ &   1.13~$+$~14 & 5 & 45 &  2 & 1.8446$^{{\rm b}}$ &   2.30~$+$~10\\
5 & 30 & 10 & 1.8996$^{{\rm b}}$ &   4.61~$+$~11 & 5 & 37 & 10 & 1.8902$^{{\rm b}}$ &   3.54~$+$~13 & 5 & 45 & 11 & 1.8841$^{{\rm b}}$ &   2.21~$+$~14\\
5 & 31 &  4 & 1.8803$^{{\rm b}}$ &   2.93~$+$~12 & 5 & 38 &  4 & 1.8701$^{{\rm b}}$ &   3.64~$+$~10 & 5 & 45 & 12 & 1.8891$^{{\rm b}}$ &   3.63~$+$~12\\
5 & 31 &  5 & 1.8822$^{{\rm b}}$ &   2.40~$+$~14 & 5 & 38 &  5 & 1.8719$^{{\rm b}}$ &   1.68~$+$~11 & 5 & 45 & 14 & 1.8953$^{{\rm b}}$ &   9.85~$+$~12\\
5 & 31 &  6 & 1.8901$^{{\rm b}}$ &   2.42~$+$~12 & 5 & 38 &  6 & 1.8798$^{{\rm b}}$ &   4.70~$+$~11 & 5 & 45 & 15 & 1.8974$^{{\rm b}}$ &   9.79~$+$~11\\
5 & 31 &  7 & 1.8910$^{{\rm b}}$ &   1.32~$+$~13 & 5 & 38 &  7 & 1.8806$^{{\rm b}}$ &   8.30~$+$~13 & 5 & 46 &  1 & 1.8367$^{{\rm b}}$ &   2.43~$+$~09\\
5 & 31 & 10 & 1.8993$^{{\rm b}}$ &   2.14~$+$~12 & 5 & 38 & 10 & 1.8889$^{{\rm b}}$ &   1.96~$+$~14 & 5 & 46 &  2 & 1.8407$^{{\rm b}}$ &   2.05~$+$~11\\
5 & 32 &  3 & 1.8784$^{{\rm b}}$ &   4.14~$+$~12 & 5 & 39 &  3 & 1.8659$^{{\rm b}}$ &   6.98~$+$~09 & 5 & 46 & 11 & 1.8801$^{{\rm b}}$ &   5.81~$+$~12\\
5 & 32 &  4 & 1.8803$^{{\rm b}}$ &   2.98~$+$~13 & 5 & 39 &  4 & 1.8678$^{{\rm b}}$ &   3.14~$+$~11 & 5 & 46 & 12 & 1.8850$^{{\rm b}}$ &   4.61~$+$~14\\
5 & 32 &  5 & 1.8822$^{{\rm b}}$ &   2.10~$+$~14 & 5 & 39 &  5 & 1.8697$^{{\rm b}}$ &   5.22~$+$~10 & 5 & 46 & 13 & 1.8861$^{{\rm b}}$ &   9.90~$+$~13\\
5 & 32 &  6 & 1.8902$^{{\rm b}}$ &   7.15~$+$~12 & 5 & 39 &  6 & 1.8775$^{{\rm b}}$ &   6.64~$+$~11 & 5 & 46 & 14 & 1.8912$^{{\rm b}}$ &   5.14~$+$~12\\
5 & 32 &  7 & 1.8910$^{{\rm b}}$ &   2.40~$+$~11 & 5 & 39 &  7 & 1.8783$^{{\rm b}}$ &   2.19~$+$~14 & 5 & 46 & 15 & 1.8933$^{{\rm b}}$ &   3.93~$+$~13\\
5 & 32 &  8 & 1.8944$^{{\rm b}}$ &   6.78~$+$~12 & 5 & 39 &  8 & 1.8817$^{{\rm b}}$ &   8.54~$+$~10 & 5 & 47 &  2 & 1.8396$^{{\rm b}}$ &   7.79~$+$~10\\
5 & 32 &  9 & 1.8988$^{{\rm b}}$ &   3.14~$+$~12 & 5 & 39 &  9 & 1.8861$^{{\rm b}}$ &   2.34~$+$~14 & 5 & 47 & 11 & 1.8789$^{{\rm b}}$ &   1.08~$+$~12\\
5 & 32 & 10 & 1.8994$^{{\rm b}}$ &   2.41~$+$~11 & 5 & 39 & 10 & 1.8866$^{{\rm b}}$ &   1.48~$+$~13 & 5 & 47 & 12 & 1.8838$^{{\rm b}}$ &   1.15~$+$~11\\
5 & 33 &  3 & 1.8741$^{{\rm b}}$ &   7.14~$+$~11 & 5 & 40 &  3 & 1.8654$^{{\rm b}}$ &   7.16~$+$~08 & 5 & 47 & 13 & 1.8849$^{{\rm b}}$ &   2.94~$+$~14\\
5 & 33 &  4 & 1.8760$^{{\rm b}}$ &   6.58~$+$~12 & 5 & 40 &  4 & 1.8674$^{{\rm b}}$ &   2.25~$+$~10 & 5 & 47 & 15 & 1.8921$^{{\rm b}}$ &   1.39~$+$~14\\
5 & 33 &  5 & 1.8779$^{{\rm b}}$ &   7.85~$+$~12 & 5 & 40 &  6 & 1.8770$^{{\rm b}}$ &   1.86~$+$~12 & 5 & 48 &  1 & 1.8334$^{{\rm b}}$ &   8.86~$+$~10\\
5 & 33 &  6 & 1.8858$^{{\rm b}}$ &   3.43~$+$~14 & 5 & 40 &  8 & 1.8812$^{{\rm b}}$ &   2.48~$+$~14 & 5 & 48 &  2 & 1.8374$^{{\rm b}}$ &   7.21~$+$~10\\
5 & 33 &  7 & 1.8866$^{{\rm b}}$ &   1.07~$+$~14 & 5 & 40 &  9 & 1.8856$^{{\rm b}}$ &   1.63~$+$~14 & 5 & 48 & 11 & 1.8766$^{{\rm b}}$ &   3.35~$+$~12\\
5 & 33 &  8 & 1.8900$^{{\rm b}}$ &   5.81~$+$~13 & 5 & 40 & 10 & 1.8861$^{{\rm b}}$ &   5.51~$+$~13 & 5 & 48 & 12 & 1.8815$^{{\rm b}}$ &   2.76~$+$~12\\
5 & 33 &  9 & 1.8944$^{{\rm b}}$ &   7.81~$+$~12 & 5 & 41 &  3 & 1.8625$^{{\rm b}}$ &   1.77~$+$~10 & 5 & 48 & 13 & 1.8826$^{{\rm b}}$ &   4.33~$+$~14\\
5 & 33 & 10 & 1.8949$^{{\rm b}}$ &   4.65~$+$~12 & 5 & 41 &  4 & 1.8645$^{{\rm b}}$ &   4.41~$+$~10 & 5 & 48 & 14 & 1.8877$^{{\rm b}}$ &   1.46~$+$~14\\
5 & 34 &  4 & 1.8746$^{{\rm b}}$ &   2.25~$+$~12 & 5 & 41 &  5 & 1.8663$^{{\rm b}}$ &   3.59~$+$~10 & 5 & 48 & 15 & 1.8898$^{{\rm b}}$ &   1.25~$+$~14\\
5 & 34 &  5 & 1.8764$^{{\rm b}}$ &   4.89~$+$~12 & 5 & 41 &  6 & 1.8741$^{{\rm b}}$ &   2.86~$+$~11 & 5 & 49 &  1 & 1.8332$^{{\rm b}}$ &   1.83~$+$~11\\
5 & 34 &  6 & 1.8843$^{{\rm b}}$ &   5.22~$+$~13 & 5 & 41 &  7 & 1.8749$^{{\rm b}}$ &   1.48~$+$~13 & 5 & 49 &  2 & 1.8372$^{{\rm b}}$ &   1.17~$+$~10\\
5 & 34 &  7 & 1.8851$^{{\rm b}}$ &   5.11~$+$~14 & 5 & 41 &  8 & 1.8783$^{{\rm b}}$ &   5.14~$+$~08 & 5 & 49 & 11 & 1.8764$^{{\rm b}}$ &   6.24~$+$~11\\
5 & 34 & 10 & 1.8934$^{{\rm b}}$ &   1.85~$+$~13 & 5 & 41 &  9 & 1.8827$^{{\rm b}}$ &   1.03~$+$~13 & 5 & 49 & 12 & 1.8813$^{{\rm b}}$ &   3.61~$+$~14\\
5 & 35 &  3 & 1.8716$^{{\rm b}}$ &   4.14~$+$~11 & 5 & 41 & 10 & 1.8832$^{{\rm b}}$ &   2.01~$+$~14 & 5 & 49 & 14 & 1.8875$^{{\rm b}}$ &   3.40~$+$~14\\
5 & 35 &  4 & 1.8735$^{{\rm b}}$ &   2.29~$+$~12 & 5 & 42 &  3 & 1.8611$^{{\rm b}}$ &   7.50~$+$~09 & 5 & 49 & 15 & 1.8896$^{{\rm b}}$ &   1.42~$+$~14\\
5 & 35 &  5 & 1.8754$^{{\rm b}}$ &   3.24~$+$~13 & 5 & 42 &  4 & 1.8630$^{{\rm b}}$ &   2.52~$+$~11 & 5 & 50 &  1 & 1.8262$^{{\rm b}}$ &   4.08~$+$~09\\
5 & 35 &  6 & 1.8832$^{{\rm b}}$ &   6.10~$+$~13 & 5 & 42 &  6 & 1.8727$^{{\rm b}}$ &   6.14~$+$~12 & 5 & 50 &  2 & 1.8301$^{{\rm b}}$ &   1.63~$+$~11\\
5 & 35 &  7 & 1.8841$^{{\rm b}}$ &   2.26~$+$~12 & 5 & 42 &  8 & 1.8768$^{{\rm b}}$ &   3.23~$+$~11 & 5 & 50 & 11 & 1.8690$^{{\rm b}}$ &   6.52~$+$~11\\
5 & 35 &  8 & 1.8874$^{{\rm b}}$ &   3.81~$+$~12 & 5 & 42 &  9 & 1.8812$^{{\rm b}}$ &   8.54~$+$~13 & 5 & 50 & 12 & 1.8740$^{{\rm b}}$ &   1.32~$+$~11\\
5 & 35 &  9 & 1.8919$^{{\rm b}}$ &   1.84~$+$~12 & 5 & 42 & 10 & 1.8817$^{{\rm b}}$ &   5.82~$+$~14 & 5 & 50 & 14 & 1.8801$^{{\rm b}}$ &   2.70~$+$~13\\
5 & 35 & 10 & 1.8924$^{{\rm b}}$ &   2.63~$+$~13 & 5 & 43 &  2 & 1.8483$^{{\rm b}}$ &   1.61~$+$~10 & 5 & 50 & 15 & 1.8821$^{{\rm b}}$ &   4.68~$+$~14\\
5 & 36 &  3 & 1.8699$^{{\rm b}}$ &   3.64~$+$~09 & 5 & 43 & 11 & 1.8880$^{{\rm b}}$ &   2.02~$+$~14 & 6 &  2 &  1 & 1354.1 &   6.54~$+$~03$^{{\rm d}}$\\
5 & 36 &  4 & 1.8719$^{{\rm b}}$ &   4.74~$+$~10 & 5 & 43 & 12 & 1.8930$^{{\rm b}}$ &   1.31~$+$~13 & 6 &  3 &  1 & 852.12 &   7.27~$-$~01$^{{\rm d}}$\\
5 & 36 &  5 & 1.8737$^{{\rm b}}$ &   6.59~$+$~10 & 5 & 43 & 13 & 1.8941$^{{\rm b}}$ &   1.66~$+$~13 & 6 &  3 &  2 & 2298.0 &   8.38~$+$~02$^{{\rm d}}$\\
5 & 36 &  6 & 1.8816$^{{\rm b}}$ &   2.15~$+$~14 & 5 & 43 & 15 & 1.9014$^{{\rm b}}$ &   7.55~$+$~10 & 6 &  4 &  1 & 408.90 &   1.28~$-$~01$^{{\rm d}}$\\
5 & 36 &  7 & 1.8824$^{{\rm b}}$ &   7.77~$+$~13 & 5 & 44 &  1 & 1.8415$^{{\rm b}}$ &   8.26~$+$~07 & 6 &  4 &  2 & 585.79 &   1.65~$+$~04$^{{\rm d}}$\\
5 & 36 &  8 & 1.8858$^{{\rm b}}$ &   1.88~$+$~14 & 5 & 44 &  2 & 1.8455$^{{\rm b}}$ &   4.17~$+$~10 & 6 &  4 &  3 & 786.12 &   1.54~$+$~04$^{{\rm d}}$\\
\hline
6 &  5 &  2 & 335.43 &   1.39~$+$~05$^{{\rm d}}$ & 6 & 17 & 11 & 123.34 &   1.86~$+$~10 & 6 & 24 &  4 & 1.9012 &   5.27~$+$~13\\
6 &  5 &  3 & 392.73 &   6.36~$+$~01$^{{\rm d}}$ & 6 & 17 & 13 & 156.24 &   1.80~$+$~10 & 6 & 24 & 16 & 1.9533 &   2.24~$+$~12\\
6 &  5 &  4 & 784.81 &   1.47~$+$~01$^{{\rm d}}$ & 6 & 17 & 15 & 210.72 &   1.14~$+$~07 & 6 & 24 & 19 & 1.9598 &   1.71~$+$~11\\
6 &  6 &  2 & 242.07 &   3.67~$+$~07$^{{\rm e}}$ & 6 & 18 &  6 & 79.773 &   2.14~$+$~08 & 6 & 25 &  1 & 1.8914 &   1.74~$+$~13\\
6 &  6 &  3 & 270.57 &   3.23~$+$~07$^{{\rm e}}$ & 6 & 18 &  7 & 103.75 &   1.48~$+$~10 & 6 & 25 &  2 & 1.8941 &   4.17~$+$~14\\
6 &  6 &  4 & 412.56 &   1.45~$+$~06$^{{\rm e}}$ & 6 & 18 &  8 & 103.82 &   2.23~$+$~10 & 6 & 25 &  3 & 1.8956 &   2.08~$+$~14\\
6 &  7 &  1 & 128.75 &   1.19~$+$~10 & 6 & 18 & 10 & 121.33 &   4.92~$+$~09 & 6 & 25 &  4 & 1.9002 &   3.90~$+$~13\\
6 &  7 &  2 & 142.28 &   8.17~$+$~08 & 6 & 18 & 11 & 122.61 &   1.22~$+$~08 & 6 & 25 &  5 & 1.9048 &   4.58~$+$~10\\
6 &  7 &  3 & 151.67 &   8.19~$+$~07 & 6 & 18 & 12 & 125.30 &   1.64~$+$~10 & 6 & 25 & 16 & 1.9522 &   3.79~$+$~12\\
6 &  7 &  4 & 187.92 &   1.89~$+$~08 & 6 & 18 & 13 & 155.08 &   1.35~$+$~10 & 6 & 25 & 17 & 1.9556 &   1.35~$+$~10\\
6 &  7 &  5 & 247.09 &   4.14~$+$~07 & 6 & 18 & 14 & 163.06 &   3.88~$+$~08 & 6 & 25 & 18 & 1.9558 &   2.91~$+$~12\\
6 &  8 &  2 & 142.15 &   9.58~$+$~09 & 6 & 18 & 15 & 208.61 &   6.44~$+$~08 & 6 & 25 & 19 & 1.9588 &   1.32~$+$~11\\
6 &  8 &  3 & 151.52 &   2.43~$+$~06 & 6 & 19 &  6 & 75.179 &   4.61~$+$~07 & 6 & 25 & 20 & 1.9677 &   5.27~$+$~09\\
6 &  8 &  4 & 187.70 &   3.92~$+$~07 & 6 & 19 &  7 & 96.116 &   3.05~$+$~06 & 6 & 26 &  2 & 1.8920 &   5.07~$+$~12\\
6 &  9 &  3 & 145.73 &   6.43~$+$~09 & 6 & 19 &  8 & 96.176 &   7.96~$+$~08 & 6 & 26 &  3 & 1.8936 &   3.10~$+$~14\\
6 &  9 &  4 & 178.90 &   9.61~$+$~08 & 6 & 19 &  9 & 98.662 &   5.60~$+$~09 & 6 & 26 &  4 & 1.8982 &   1.70~$+$~14\\
6 & 10 &  2 & 118.70 &   2.28~$+$~10 & 6 & 19 & 11 & 112.09 &   1.21~$+$~09 & 6 & 26 & 16 & 1.9500 &   1.21~$+$~12\\
6 & 11 &  1 & 108.12 &   4.53~$+$~09 & 6 & 19 & 12 & 114.33 &   2.54~$+$~09 & 6 & 26 & 18 & 1.9536 &   8.85~$+$~11\\
6 & 11 &  2 & 117.50 &   1.65~$+$~10 & 6 & 19 & 13 & 138.61 &   4.77~$+$~06 & 6 & 26 & 19 & 1.9566 &   2.88~$+$~12\\
6 & 11 &  3 & 123.83 &   2.81~$+$~09 & 6 & 19 & 14 & 144.96 &   3.31~$+$~10 & 6 & 27 &  2 & 1.8914$^{{\rm b}}$ &   2.23~$+$~14\\
6 & 11 &  4 & 146.98 &   2.63~$+$~08 & 6 & 19 & 15 & 179.87 &   4.66~$+$~09 & 6 & 27 & 18 & 1.9530$^{{\rm b}}$ &   2.85~$+$~12\\
6 & 11 &  5 & 180.85 &   1.52~$+$~08 & 6 & 20 &  7 & 78.647 &   2.36~$+$~08 & 6 & 28 &  1 & 1.8883$^{{\rm b}}$ &   1.99~$+$~12\\
6 & 12 &  2 & 115.13 &   4.50~$+$~08 & 6 & 20 & 11 & 89.025 &   4.13~$+$~09 & 6 & 28 &  2 & 1.8910$^{{\rm b}}$ &   1.73~$+$~13\\
6 & 12 &  3 & 121.20 &   2.12~$+$~10 & 6 & 20 & 13 & 104.98 &   5.14~$+$~09 & 6 & 28 &  3 & 1.8925$^{{\rm b}}$ &   2.48~$+$~14\\
6 & 12 &  4 & 143.29 &   1.18~$+$~08 & 6 & 20 & 15 & 127.06 &   7.87~$+$~10 & 6 & 28 &  4 & 1.8971$^{{\rm b}}$ &   1.34~$+$~12\\
6 & 13 &  1 & 91.268 &   9.59~$+$~09 & 6 & 21 &  2 & 1.9048$^{{\rm b}}$ &   7.47~$+$~12 & 6 & 28 &  5 & 1.9017$^{{\rm b}}$ &   4.50~$+$~13\\
6 & 13 &  2 & 97.865 &   2.60~$+$~10 & 6 & 21 &  3 & 1.9064$^{{\rm b}}$ &   7.03~$+$~12 & 6 & 28 & 16 & 1.9489$^{{\rm b}}$ &   1.86~$+$~10\\
6 & 13 &  3 & 102.22 &   6.28~$+$~10 & 6 & 21 &  4 & 1.9110$^{{\rm b}}$ &   2.49~$+$~11 & 6 & 28 & 17 & 1.9523$^{{\rm b}}$ &   2.27~$+$~12\\
6 & 13 &  4 & 117.49 &   3.01~$+$~08 & 6 & 21 & 16 & 1.9636$^{{\rm b}}$ &   9.12~$+$~10 & 6 & 28 & 18 & 1.9525$^{{\rm b}}$ &   1.06~$+$~12\\
6 & 13 &  5 & 138.18 &   6.39~$+$~08 & 6 & 21 & 18 & 1.9673$^{{\rm b}}$ &   3.65~$+$~09 & 6 & 28 & 19 & 1.9554$^{{\rm b}}$ &   6.29~$+$~11\\
6 & 14 &  2 & 94.932 &   3.95~$+$~08 & 6 & 21 & 19 & 1.9702$^{{\rm b}}$ &   8.04~$+$~08 & 6 & 28 & 20 & 1.9643$^{{\rm b}}$ &   2.18~$+$~09\\
6 & 14 &  3 & 99.021 &   7.50~$+$~09 & 6 & 22 &  1 & 1.8947 &   2.72~$+$~14 & 6 & 29 &  2 & 1.8881 &   4.71~$+$~12\\
6 & 14 &  4 & 113.29 &   4.66~$+$~10 & 6 & 22 &  2 & 1.8974 &   4.05~$+$~12 & 6 & 29 &  3 & 1.8897 &   7.42~$+$~12\\
6 & 15 &  1 & 79.293 &   2.89~$+$~07 & 6 & 22 &  3 & 1.8990 &   7.98~$+$~13 & 6 & 29 &  4 & 1.8942 &   3.53~$+$~14\\
6 & 15 &  2 & 84.225 &   4.81~$+$~09 & 6 & 22 &  4 & 1.9036 &   3.63~$+$~12 & 6 & 29 & 16 & 1.9459 &   1.80~$+$~11\\
6 & 15 &  3 & 87.429 &   2.15~$+$~08 & 6 & 22 &  5 & 1.9082 &   2.52~$+$~10 & 6 & 29 & 18 & 1.9495 &   1.32~$+$~12\\
6 & 15 &  4 & 98.369 &   6.79~$+$~10 & 6 & 22 & 16 & 1.9557 &   2.31~$+$~12 & 6 & 29 & 19 & 1.9524 &   3.32~$+$~12\\
6 & 15 &  5 & 112.47 &   1.77~$+$~10 & 6 & 22 & 17 & 1.9592 &   8.20~$+$~11 & 6 & 30 &  1 & 1.8831 &   8.27~$+$~10\\
6 & 16 &  6 & 86.255 &   1.34~$+$~09 & 6 & 22 & 18 & 1.9594 &   1.34~$+$~11 & 6 & 30 &  2 & 1.8857 &   1.27~$+$~12\\
6 & 16 &  7 & 114.99 &   3.44~$+$~09 & 6 & 22 & 19 & 1.9623 &   1.62~$+$~10 & 6 & 30 &  3 & 1.8873 &   8.97~$+$~07\\
6 & 16 &  8 & 115.08 &   1.36~$+$~10 & 6 & 22 & 20 & 1.9712 &   7.29~$+$~09 & 6 & 30 &  4 & 1.8918 &   3.85~$+$~14\\
6 & 16 &  9 & 118.66 &   2.94~$+$~10 & 6 & 23 &  2 & 1.8970 &   2.00~$+$~14 & 6 & 30 &  5 & 1.8964 &   2.42~$+$~14\\
6 & 16 & 11 & 138.62 &   3.84~$+$~09 & 6 & 23 &  3 & 1.8985 &   4.26~$+$~12 & 6 & 30 & 16 & 1.9433 &   1.94~$+$~10\\
6 & 16 & 12 & 142.07 &   3.77~$+$~09 & 6 & 23 &  4 & 1.9031 &   2.42~$+$~13 & 6 & 30 & 17 & 1.9467 &   9.12~$+$~10\\
6 & 16 & 13 & 181.61 &   6.31~$+$~09 & 6 & 23 & 16 & 1.9553 &   1.90~$+$~12 & 6 & 30 & 18 & 1.9469 &   2.91~$+$~11\\
6 & 16 & 14 & 192.66 &   7.76~$+$~08 & 6 & 23 & 18 & 1.9589 &   4.61~$+$~11 & 6 & 30 & 19 & 1.9498 &   5.02~$+$~12\\
6 & 16 & 15 & 259.63 &   8.63~$+$~07 & 6 & 23 & 19 & 1.9618 &   8.59~$+$~09 & 6 & 30 & 20 & 1.9586 &   1.93~$+$~12\\
6 & 17 &  7 & 104.27 &   3.62~$+$~10 & 6 & 24 &  3 & 1.8966 &   1.62~$+$~14 & 6 & 31 &  6 & 1.8995$^{{\rm b}}$ &   2.10~$+$~14\\
\hline
6 & 31 &  8 & 1.9100$^{{\rm b}}$ &   3.52~$+$~12 & 6 & 37 &  9 & 1.8959$^{{\rm b}}$ &   2.21~$+$~14 & 6 & 43 & 14 & 1.8987$^{{\rm b}}$ &   1.81~$+$~14\\
6 & 31 &  9 & 1.9110$^{{\rm b}}$ &   6.08~$+$~12 & 6 & 37 & 11 & 1.9002$^{{\rm b}}$ &   1.65~$+$~14 & 6 & 43 & 15 & 1.9036$^{{\rm b}}$ &   1.24~$+$~14\\
6 & 31 & 12 & 1.9160$^{{\rm b}}$ &   2.82~$+$~11 & 6 & 37 & 12 & 1.9009$^{{\rm b}}$ &   3.86~$+$~13 & 6 & 44 &  6 & 1.8751$^{{\rm b}}$ &   8.50~$+$~09\\
6 & 31 & 14 & 1.9229$^{{\rm b}}$ &   8.13~$+$~08 & 6 & 37 & 13 & 1.9064$^{{\rm b}}$ &   6.99~$+$~12 & 6 & 44 &  7 & 1.8853$^{{\rm b}}$ &   2.78~$+$~11\\
6 & 32 &  6 & 1.8968$^{{\rm b}}$ &   2.17~$+$~14 & 6 & 37 & 14 & 1.9076$^{{\rm b}}$ &   3.13~$+$~12 & 6 & 44 &  8 & 1.8853$^{{\rm b}}$ &   1.09~$+$~12\\
6 & 32 &  7 & 1.9073$^{{\rm b}}$ &   6.23~$+$~11 & 6 & 37 & 15 & 1.9125$^{{\rm b}}$ &   1.71~$+$~12 & 6 & 44 & 10 & 1.8903$^{{\rm b}}$ &   7.20~$+$~11\\
6 & 32 &  8 & 1.9073$^{{\rm b}}$ &   1.76~$+$~12 & 6 & 38 &  6 & 1.8836$^{{\rm b}}$ &   3.37~$+$~11 & 6 & 44 & 11 & 1.8906$^{{\rm b}}$ &   1.80~$+$~11\\
6 & 32 &  9 & 1.9083$^{{\rm b}}$ &   1.27~$+$~12 & 6 & 38 &  7 & 1.8939$^{{\rm b}}$ &   1.79~$+$~12 & 6 & 44 & 12 & 1.8912$^{{\rm b}}$ &   9.76~$+$~13\\
6 & 32 & 11 & 1.9127$^{{\rm b}}$ &   4.14~$+$~11 & 6 & 38 &  8 & 1.8939$^{{\rm b}}$ &   2.98~$+$~14 & 6 & 44 & 13 & 1.8967$^{{\rm b}}$ &   2.12~$+$~14\\
6 & 32 & 12 & 1.9133$^{{\rm b}}$ &   2.31~$+$~12 & 6 & 38 & 10 & 1.8989$^{{\rm b}}$ &   2.01~$+$~14 & 6 & 44 & 14 & 1.8979$^{{\rm b}}$ &   4.41~$+$~13\\
6 & 32 & 13 & 1.9189$^{{\rm b}}$ &   2.92~$+$~11 & 6 & 38 & 11 & 1.8992$^{{\rm b}}$ &   2.59~$+$~13 & 6 & 44 & 15 & 1.9027$^{{\rm b}}$ &   9.36~$+$~11\\
6 & 32 & 14 & 1.9201$^{{\rm b}}$ &   3.07~$+$~10 & 6 & 38 & 12 & 1.8998$^{{\rm b}}$ &   1.32~$+$~14 & 6 & 45 &  6 & 1.8711$^{{\rm b}}$ &   1.12~$+$~07\\
6 & 32 & 15 & 1.9251$^{{\rm b}}$ &   1.53~$+$~10 & 6 & 38 & 13 & 1.9054$^{{\rm b}}$ &   1.10~$+$~13 & 6 & 45 &  7 & 1.8813$^{{\rm b}}$ &   2.02~$+$~10\\
6 & 33 &  6 & 1.8957$^{{\rm b}}$ &   2.18~$+$~14 & 6 & 38 & 14 & 1.9065$^{{\rm b}}$ &   9.32~$+$~12 & 6 & 45 &  8 & 1.8813$^{{\rm b}}$ &   1.56~$+$~12\\
6 & 33 &  7 & 1.9061$^{{\rm b}}$ &   1.88~$+$~11 & 6 & 38 & 15 & 1.9114$^{{\rm b}}$ &   7.63~$+$~11 & 6 & 45 & 10 & 1.8862$^{{\rm b}}$ &   4.27~$+$~11\\
6 & 33 &  8 & 1.9061$^{{\rm b}}$ &   3.21~$+$~11 & 6 & 39 &  7 & 1.8937$^{{\rm b}}$ &   2.38~$+$~14 & 6 & 45 & 11 & 1.8865$^{{\rm b}}$ &   4.28~$+$~11\\
6 & 33 & 10 & 1.9112$^{{\rm b}}$ &   1.60~$+$~12 & 6 & 39 & 11 & 1.8990$^{{\rm b}}$ &   5.52~$+$~14 & 6 & 45 & 12 & 1.8871$^{{\rm b}}$ &   3.91~$+$~11\\
6 & 33 & 11 & 1.9115$^{{\rm b}}$ &   3.32~$+$~12 & 6 & 39 & 13 & 1.9052$^{{\rm b}}$ &   4.36~$+$~13 & 6 & 45 & 13 & 1.8926$^{{\rm b}}$ &   8.00~$+$~12\\
6 & 33 & 12 & 1.9122$^{{\rm b}}$ &   6.97~$+$~11 & 6 & 39 & 15 & 1.9112$^{{\rm b}}$ &   1.58~$+$~12 & 6 & 45 & 14 & 1.8938$^{{\rm b}}$ &   4.92~$+$~14\\
6 & 33 & 13 & 1.9178$^{{\rm b}}$ &   3.09~$+$~11 & 6 & 40 &  6 & 1.8790$^{{\rm b}}$ &   2.85~$+$~11 & 6 & 45 & 15 & 1.8986$^{{\rm b}}$ &   2.86~$+$~14\\
6 & 33 & 14 & 1.9189$^{{\rm b}}$ &   9.44~$+$~09 & 6 & 40 &  7 & 1.8892$^{{\rm b}}$ &   6.49~$+$~11 & 6 & 46 &  7 & 1.8790$^{{\rm b}}$ &   2.82~$+$~11\\
6 & 33 & 15 & 1.9239$^{{\rm b}}$ &   1.46~$+$~10 & 6 & 40 &  8 & 1.8893$^{{\rm b}}$ &   7.89~$+$~12 & 6 & 46 & 11 & 1.8843$^{{\rm b}}$ &   3.20~$+$~11\\
6 & 34 &  6 & 1.8868$^{{\rm b}}$ &   3.43~$+$~12 & 6 & 40 & 10 & 1.8942$^{{\rm b}}$ &   1.28~$+$~13 & 6 & 46 & 13 & 1.8903$^{{\rm b}}$ &   2.73~$+$~11\\
6 & 34 &  7 & 1.8972$^{{\rm b}}$ &   4.23~$+$~13 & 6 & 40 & 11 & 1.8945$^{{\rm b}}$ &   9.71~$+$~13 & 6 & 46 & 15 & 1.8963$^{{\rm b}}$ &   3.90~$+$~14\\
6 & 34 &  8 & 1.8972$^{{\rm b}}$ &   4.10~$+$~14 & 6 & 40 & 12 & 1.8952$^{{\rm b}}$ &   2.81~$+$~14 & 6 & 47 &  2 & 1.8433$^{{\rm b}}$ &   1.60~$+$~10\\
6 & 34 &  9 & 1.8981$^{{\rm b}}$ &   2.18~$+$~14 & 6 & 40 & 13 & 1.9007$^{{\rm b}}$ &   7.75~$+$~12 & 6 & 47 &  3 & 1.8448$^{{\rm b}}$ &   4.52~$+$~10\\
6 & 34 & 11 & 1.9025$^{{\rm b}}$ &   2.28~$+$~12 & 6 & 40 & 14 & 1.9018$^{{\rm b}}$ &   1.32~$+$~12 & 6 & 47 &  4 & 1.8491$^{{\rm b}}$ &   2.81~$+$~10\\
6 & 34 & 12 & 1.9031$^{{\rm b}}$ &   6.37~$+$~13 & 6 & 40 & 15 & 1.9067$^{{\rm b}}$ &   1.77~$+$~13 & 6 & 47 & 16 & 1.8983$^{{\rm b}}$ &   2.86~$+$~14\\
6 & 34 & 13 & 1.9087$^{{\rm b}}$ &   3.08~$+$~12 & 6 & 41 &  6 & 1.8785$^{{\rm b}}$ &   1.47~$+$~10 & 6 & 47 & 18 & 1.9017$^{{\rm b}}$ &   1.06~$+$~14\\
6 & 34 & 14 & 1.9099$^{{\rm b}}$ &   1.99~$+$~12 & 6 & 41 &  7 & 1.8888$^{{\rm b}}$ &   3.67~$+$~11 & 6 & 47 & 19 & 1.9045$^{{\rm b}}$ &   3.40~$+$~13\\
6 & 34 & 15 & 1.9148$^{{\rm b}}$ &   2.19~$+$~11 & 6 & 41 &  8 & 1.8888$^{{\rm b}}$ &   1.30~$+$~12 & 6 & 48 &  1 & 1.8386$^{{\rm b}}$ &   2.31~$+$~10\\
6 & 35 &  6 & 1.8859$^{{\rm b}}$ &   3.02~$+$~11 & 6 & 41 &  9 & 1.8898$^{{\rm b}}$ &   7.59~$+$~13 & 6 & 48 &  2 & 1.8411$^{{\rm b}}$ &   1.19~$+$~10\\
6 & 35 &  7 & 1.8962$^{{\rm b}}$ &   5.03~$+$~14 & 6 & 41 & 11 & 1.8941$^{{\rm b}}$ &   8.09~$+$~12 & 6 & 48 &  3 & 1.8426$^{{\rm b}}$ &   8.39~$+$~09\\
6 & 35 &  8 & 1.8962$^{{\rm b}}$ &   3.36~$+$~12 & 6 & 41 & 12 & 1.8947$^{{\rm b}}$ &   4.18~$+$~11 & 6 & 48 &  4 & 1.8469$^{{\rm b}}$ &   6.31~$+$~10\\
6 & 35 & 10 & 1.9013$^{{\rm b}}$ &   1.38~$+$~12 & 6 & 41 & 13 & 1.9003$^{{\rm b}}$ &   1.60~$+$~14 & 6 & 48 &  5 & 1.8513$^{{\rm b}}$ &   1.73~$+$~10\\
6 & 35 & 11 & 1.9016$^{{\rm b}}$ &   1.85~$+$~13 & 6 & 41 & 14 & 1.9014$^{{\rm b}}$ &   1.32~$+$~14 & 6 & 48 & 16 & 1.8960$^{{\rm b}}$ &   2.25~$+$~14\\
6 & 35 & 12 & 1.9022$^{{\rm b}}$ &   6.09~$+$~13 & 6 & 41 & 15 & 1.9063$^{{\rm b}}$ &   6.30~$+$~11 & 6 & 48 & 17 & 1.8992$^{{\rm b}}$ &   1.43~$+$~14\\
6 & 35 & 13 & 1.9078$^{{\rm b}}$ &   6.61~$+$~12 & 6 & 42 &  7 & 1.8862$^{{\rm b}}$ &   3.30~$+$~12 & 6 & 48 & 18 & 1.8994$^{{\rm b}}$ &   9.21~$+$~13\\
6 & 35 & 14 & 1.9089$^{{\rm b}}$ &   1.11~$+$~12 & 6 & 42 & 11 & 1.8915$^{{\rm b}}$ &   8.47~$+$~13 & 6 & 48 & 19 & 1.9021$^{{\rm b}}$ &   2.83~$+$~13\\
6 & 35 & 15 & 1.9138$^{{\rm b}}$ &   1.00~$+$~10 & 6 & 42 & 13 & 1.8976$^{{\rm b}}$ &   2.07~$+$~14 & 6 & 48 & 20 & 1.9105$^{{\rm b}}$ &   3.51~$+$~08\\
6 & 36 &  6 & 1.8850$^{{\rm b}}$ &   1.70~$+$~11 & 6 & 42 & 15 & 1.9036$^{{\rm b}}$ &   2.53~$+$~13 & 6 & 49 &  2 & 1.8386$^{{\rm b}}$ &   5.84~$+$~10\\
6 & 36 &  8 & 1.8954$^{{\rm b}}$ &   9.80~$+$~11 & 6 & 43 &  6 & 1.8759$^{{\rm b}}$ &   1.14~$+$~10 & 6 & 49 & 18 & 1.8968$^{{\rm b}}$ &   4.28~$+$~14\\
6 & 36 &  9 & 1.8964$^{{\rm b}}$ &   2.84~$+$~14 & 6 & 43 &  7 & 1.8862$^{{\rm b}}$ &   3.74~$+$~11 & 6 & 50 &  1 & 1.8333$^{{\rm b}}$ &   1.59~$+$~09\\
6 & 36 & 12 & 1.9014$^{{\rm b}}$ &   1.33~$+$~14 & 6 & 43 &  8 & 1.8862$^{{\rm b}}$ &   5.07~$+$~11 & 6 & 50 &  2 & 1.8358$^{{\rm b}}$ &   2.12~$+$~09\\
6 & 36 & 14 & 1.9081$^{{\rm b}}$ &   1.09~$+$~13 & 6 & 43 &  9 & 1.8871$^{{\rm b}}$ &   1.29~$+$~12 & 6 & 50 &  3 & 1.8373$^{{\rm b}}$ &   3.59~$+$~10\\
6 & 37 &  6 & 1.8846$^{{\rm b}}$ &   1.12~$+$~12 & 6 & 43 & 11 & 1.8915$^{{\rm b}}$ &   1.95~$+$~12 & 6 & 50 &  4 & 1.8416$^{{\rm b}}$ &   7.42~$+$~10\\
6 & 37 &  7 & 1.8949$^{{\rm b}}$ &   1.73~$+$~10 & 6 & 43 & 12 & 1.8921$^{{\rm b}}$ &   2.01~$+$~13 & 6 & 50 &  5 & 1.8459$^{{\rm b}}$ &   1.22~$+$~11\\
6 & 37 &  8 & 1.8949$^{{\rm b}}$ &   5.19~$+$~13 & 6 & 43 & 13 & 1.8976$^{{\rm b}}$ &   2.40~$+$~13 & 6 & 50 & 16 & 1.8904$^{{\rm b}}$ &   1.25~$+$~13\\
\hline
6 & 50 & 17 & 1.8936$^{{\rm b}}$ &   1.63~$+$~10 & 7 & 13 &  5 & 98.357 &   9.12~$+$~10 & 7 & 21 &  5 & 1.9094 &   5.13~$+$~13\\
6 & 50 & 18 & 1.8938$^{{\rm b}}$ &   1.47~$+$~13 & 7 & 14 &  6 & 83.224 &   2.77~$+$~09 & 7 & 21 & 14 & 1.9708 &   3.56~$+$~12\\
6 & 50 & 19 & 1.8965$^{{\rm b}}$ &   6.38~$+$~14 & 7 & 14 &  7 & 88.208 &   1.50~$+$~09 & 7 & 21 & 15 & 1.9749 &   7.65~$+$~11\\
6 & 50 & 20 & 1.9049$^{{\rm b}}$ &   1.47~$+$~14 & 7 & 14 &  8 & 89.942 &   4.63~$+$~08 & 7 & 22 &  1 & 1.8974 &   3.33~$+$~12\\
7 &  2 &  1 & 721.40 &   1.49~$+$~04$^{{\rm f}}$ & 7 & 14 &  9 & 109.68 &   1.59~$+$~10 & 7 & 22 &  2 & 1.9024 &   2.96~$+$~12\\
7 &  3 &  1 & 567.76 &   1.44~$+$~03$^{{\rm f}}$ & 7 & 14 & 10 & 111.61 &   4.04~$+$~10 & 7 & 22 &  3 & 1.9037 &   3.92~$+$~14\\
7 &  3 &  2 & 2665.2 &   4.24~$+$~02$^{{\rm f}}$ & 7 & 14 & 11 & 131.74 &   8.55~$+$~09 & 7 & 22 &  4 & 1.9068 &   1.35~$+$~14\\
7 &  4 &  1 & 384.22 &   3.16~$+$~04$^{{\rm f}}$ & 7 & 14 & 12 & 140.47 &   2.85~$+$~10 & 7 & 22 &  5 & 1.9091 &   1.12~$+$~14\\
7 &  4 &  2 & 822.03 &   5.27~$+$~03$^{{\rm f}}$ & 7 & 14 & 13 & 162.79 &   1.52~$+$~09 & 7 & 22 & 14 & 1.9704 &   1.47~$+$~12\\
7 &  4 &  3 & 1188.5 &   7.29~$-$~01$^{{\rm f}}$ & 7 & 15 &  7 & 80.557 &   2.63~$+$~08 & 7 & 22 & 15 & 1.9746 &   2.08~$+$~12\\
7 &  5 &  1 & 309.27 &   2.42~$+$~04$^{{\rm f}}$ & 7 & 15 &  8 & 82.000 &   8.37~$+$~08 & 7 & 23 &  1 & 1.8930 &   1.47~$+$~11\\
7 &  5 &  2 & 541.36 &   4.53~$+$~04$^{{\rm f}}$ & 7 & 15 &  9 & 98.095 &   4.30~$+$~10 & 7 & 23 &  2 & 1.8980 &   8.78~$+$~12\\
7 &  5 &  3 & 679.30 &   1.19~$+$~04$^{{\rm f}}$ & 7 & 15 & 11 & 115.38 &   2.45~$+$~09 & 7 & 23 &  4 & 1.9023 &   7.78~$+$~10\\
7 &  5 &  4 & 1585.5 &   1.61~$+$~03$^{{\rm f}}$ & 7 & 15 & 12 & 122.02 &   3.35~$+$~10 & 7 & 23 &  5 & 1.9046 &   5.58~$+$~14\\
7 &  6 &  1 & 132.85 &   1.26~$+$~10 & 7 & 15 & 13 & 138.51 &   2.95~$+$~10 & 7 & 23 & 14 & 1.9657 &   6.01~$+$~10\\
7 &  6 &  2 & 162.84 &   5.42~$+$~08 & 7 & 16 &  1 & 1.9085 &   1.89~$+$~14 & 7 & 23 & 15 & 1.9699 &   3.94~$+$~12\\
7 &  6 &  3 & 173.43 &   2.46~$+$~08 & 7 & 16 &  2 & 1.9135 &   1.52~$+$~13 & 7 & 24 &  6 & 1.9085$^{{\rm b}}$ &   2.85~$+$~14\\
7 &  6 &  5 & 232.89 &   2.33~$+$~07 & 7 & 16 &  3 & 1.9149 &   1.44~$+$~13 & 7 & 24 &  7 & 1.9110$^{{\rm b}}$ &   1.23~$+$~14\\
7 &  7 &  1 & 121.86 &   1.76~$+$~10 & 7 & 16 &  5 & 1.9203 &   3.35~$+$~11 & 7 & 24 &  9 & 1.9191$^{{\rm b}}$ &   2.00~$+$~12\\
7 &  7 &  2 & 146.63 &   7.10~$+$~07 & 7 & 16 & 14 & 1.9824 &   5.33~$+$~10 & 7 & 24 & 10 & 1.9197$^{{\rm b}}$ &   8.70~$+$~12\\
7 &  7 &  3 & 155.16 &   5.11~$+$~07 & 7 & 17 &  1 & 1.9057 &   2.23~$+$~14 & 7 & 24 & 12 & 1.9265$^{{\rm b}}$ &   3.61~$+$~11\\
7 &  7 &  4 & 178.46 &   3.33~$+$~06 & 7 & 17 &  2 & 1.9107 &   2.67~$+$~12 & 7 & 25 &  6 & 1.9063$^{{\rm b}}$ &   1.97~$+$~14\\
7 &  7 &  5 & 201.09 &   1.60~$+$~08 & 7 & 17 &  3 & 1.9121 &   9.14~$+$~12 & 7 & 25 &  7 & 1.9087$^{{\rm b}}$ &   5.22~$+$~13\\
7 &  8 &  1 & 118.70 &   1.97~$+$~10 & 7 & 17 &  4 & 1.9152 &   1.78~$+$~11 & 7 & 25 &  8 & 1.9095$^{{\rm b}}$ &   1.72~$+$~14\\
7 &  8 &  2 & 142.07 &   2.74~$+$~08 & 7 & 17 &  5 & 1.9175 &   3.80~$+$~12 & 7 & 25 &  9 & 1.9169$^{{\rm b}}$ &   4.09~$+$~12\\
7 &  8 &  4 & 171.76 &   2.22~$+$~08 & 7 & 17 & 14 & 1.9794 &   1.64~$+$~11 & 7 & 25 & 10 & 1.9174$^{{\rm b}}$ &   3.03~$+$~12\\
7 &  8 &  5 & 192.63 &   8.61~$+$~06 & 7 & 17 & 15 & 1.9836 &   2.79~$+$~09 & 7 & 25 & 11 & 1.9225$^{{\rm b}}$ &   1.20~$+$~12\\
7 &  9 &  1 & 95.917 &   1.46~$+$~09 & 7 & 18 &  1 & 1.9051 &   2.12~$+$~14 & 7 & 25 & 12 & 1.9242$^{{\rm b}}$ &   8.86~$+$~11\\
7 &  9 &  2 & 110.63 &   4.02~$+$~10 & 7 & 18 &  2 & 1.9101 &   1.01~$+$~12 & 7 & 25 & 13 & 1.9278$^{{\rm b}}$ &   4.04~$+$~10\\
7 &  9 &  3 & 115.41 &   3.99~$+$~07 & 7 & 18 &  4 & 1.9146 &   1.97~$+$~13 & 7 & 26 &  7 & 1.9068$^{{\rm b}}$ &   3.53~$+$~14\\
7 &  9 &  4 & 127.83 &   2.80~$+$~09 & 7 & 18 &  5 & 1.9169 &   2.57~$+$~12 & 7 & 26 &  8 & 1.9076$^{{\rm b}}$ &   7.09~$+$~13\\
7 &  9 &  5 & 139.04 &   5.14~$+$~08 & 7 & 18 & 14 & 1.9788 &   1.75~$+$~11 & 7 & 26 &  9 & 1.9149$^{{\rm b}}$ &   2.08~$+$~12\\
7 & 10 &  1 & 94.486 &   2.07~$+$~07 & 7 & 18 & 15 & 1.9830 &   4.63~$+$~10 & 7 & 26 & 11 & 1.9205$^{{\rm b}}$ &   8.78~$+$~10\\
7 & 10 &  2 & 108.73 &   8.76~$+$~06 & 7 & 19 &  1 & 1.9005 &   1.04~$+$~13 & 7 & 26 & 12 & 1.9223$^{{\rm b}}$ &   1.46~$+$~12\\
7 & 10 &  3 & 113.35 &   3.07~$+$~10 & 7 & 19 &  2 & 1.9055 &   4.50~$+$~14 & 7 & 26 & 13 & 1.9259$^{{\rm b}}$ &   1.02~$+$~12\\
7 & 10 &  5 & 136.05 &   5.67~$+$~09 & 7 & 19 &  3 & 1.9069 &   1.15~$+$~14 & 7 & 27 &  6 & 1.8977$^{{\rm b}}$ &   4.93~$+$~12\\
7 & 11 &  1 & 83.664 &   1.60~$+$~09 & 7 & 19 &  4 & 1.9100 &   2.70~$+$~12 & 7 & 27 &  7 & 1.9002$^{{\rm b}}$ &   5.83~$+$~12\\
7 & 11 &  2 & 94.640 &   4.30~$+$~10 & 7 & 19 &  5 & 1.9123 &   1.79~$+$~13 & 7 & 27 &  8 & 1.9009$^{{\rm b}}$ &   6.24~$+$~11\\
7 & 11 &  4 & 106.95 &   3.58~$+$~10 & 7 & 19 & 14 & 1.9738 &   2.97~$+$~12 & 7 & 27 &  9 & 1.9082$^{{\rm b}}$ &   8.19~$+$~13\\
7 & 11 &  5 & 114.69 &   2.50~$+$~09 & 7 & 19 & 15 & 1.9780 &   4.16~$+$~10 & 7 & 27 & 10 & 1.9088$^{{\rm b}}$ &   4.76~$+$~14\\
7 & 12 &  1 & 80.487 &   3.92~$+$~09 & 7 & 20 &  1 & 1.8992 &   2.65~$+$~11 & 7 & 27 & 11 & 1.9138$^{{\rm b}}$ &   1.21~$+$~14\\
7 & 12 &  2 & 90.595 &   1.42~$+$~10 & 7 & 20 &  2 & 1.9042 &   1.53~$+$~12 & 7 & 27 & 12 & 1.9155$^{{\rm b}}$ &   5.32~$+$~13\\
7 & 12 &  3 & 93.782 &   9.68~$+$~10 & 7 & 20 &  3 & 1.9056 &   2.81~$+$~14 & 7 & 27 & 13 & 1.9191$^{{\rm b}}$ &   2.19~$+$~12\\
7 & 12 &  4 & 101.82 &   8.57~$+$~09 & 7 & 20 &  5 & 1.9109 &   1.26~$+$~14 & 7 & 28 &  7 & 1.8989$^{{\rm b}}$ &   9.92~$+$~10\\
7 & 12 &  5 & 108.80 &   9.42~$+$~09 & 7 & 20 & 14 & 1.9724 &   1.83~$+$~12 & 7 & 28 &  8 & 1.8997$^{{\rm b}}$ &   2.27~$+$~12\\
7 & 13 &  1 & 74.625 &   1.10~$+$~08 & 7 & 21 &  1 & 1.8977 &   2.11~$+$~10 & 7 & 28 &  9 & 1.9069$^{{\rm b}}$ &   6.66~$+$~14\\
7 & 13 &  2 & 83.235 &   2.86~$+$~10 & 7 & 21 &  2 & 1.9027 &   3.02~$+$~14 & 7 & 28 & 11 & 1.9125$^{{\rm b}}$ &   9.82~$+$~13\\
7 & 13 &  4 & 92.612 &   3.86~$+$~09 & 7 & 21 &  4 & 1.9071 &   3.31~$+$~14 & 7 & 28 & 12 & 1.9142$^{{\rm b}}$ &   4.06~$+$~13\\
\hline
7 & 28 & 13 & 1.9178$^{{\rm b}}$ &   3.80~$+$~13 & 8 &  4 &  2 & 1068.4 &   7.07~$-$~02$^{{\rm d}}$ & 8 & 12 &  2 & 1.9182$^{{\rm b}}$ &   1.33~$+$~14\\
7 & 29 &  6 & 1.8920$^{{\rm b}}$ &   1.16~$+$~11 & 8 &  4 &  3 & 1259.3 &   6.69~$+$~02$^{{\rm d}}$ & 8 & 12 &  3 & 1.9187$^{{\rm b}}$ &   8.68~$+$~13\\
7 & 29 &  7 & 1.8945$^{{\rm b}}$ &   6.22~$+$~09 & 8 &  5 &  1 & 307.56 &   9.71~$+$~00$^{{\rm d}}$ & 8 & 12 &  4 & 1.9216$^{{\rm b}}$ &   3.05~$+$~13\\
7 & 29 &  8 & 1.8953$^{{\rm b}}$ &   1.13~$+$~11 & 8 &  5 &  3 & 424.27 &   1.43~$+$~05$^{{\rm d}}$ & 8 & 12 &  5 & 1.9274$^{{\rm b}}$ &   9.41~$+$~11\\
7 & 29 &  9 & 1.9025$^{{\rm b}}$ &   9.72~$+$~09 & 8 &  5 &  4 & 639.84 &   4.90~$+$~01$^{{\rm d}}$ & 8 & 12 & 10 & 1.9971$^{{\rm b}}$ &   3.00~$+$~11\\
7 & 29 & 10 & 1.9030$^{{\rm b}}$ &   1.50~$+$~14 & 8 &  6 &  1 & 108.36 &   3.59~$+$~10 & 8 & 13 &  3 & 1.9162$^{{\rm b}}$ &   4.04~$+$~14\\
7 & 29 & 11 & 1.9080$^{{\rm b}}$ &   2.25~$+$~12 & 8 &  6 &  3 & 119.98 &   9.65~$+$~09 & 8 & 14 &  1 & 1.9100$^{{\rm b}}$ &   8.80~$+$~12\\
7 & 29 & 12 & 1.9097$^{{\rm b}}$ &   2.82~$+$~14 & 8 &  6 &  4 & 132.62 &   2.00~$+$~09 & 8 & 14 &  2 & 1.9128$^{{\rm b}}$ &   7.55~$+$~11\\
7 & 29 & 13 & 1.9133$^{{\rm b}}$ &   8.54~$+$~13 & 8 &  7 &  1 & 101.55 &   2.93~$+$~10 & 8 & 14 &  3 & 1.9133$^{{\rm b}}$ &   1.40~$+$~13\\
7 & 30 &  7 & 1.8927$^{{\rm b}}$ &   1.37~$+$~10 & 8 &  7 &  2 & 109.95 &   1.50~$+$~10 & 8 & 14 &  4 & 1.9162$^{{\rm b}}$ &   6.05~$+$~14\\
7 & 30 &  8 & 1.8935$^{{\rm b}}$ &   3.43~$+$~11 & 8 &  7 &  3 & 111.69 &   1.16~$+$~10 & 8 & 14 &  5 & 1.9219$^{{\rm b}}$ &   1.31~$+$~14\\
7 & 30 &  9 & 1.9007$^{{\rm b}}$ &   9.96~$+$~12 & 8 &  7 &  4 & 122.57 &   1.71~$+$~07 & 8 & 14 & 10 & 1.9912$^{{\rm b}}$ &   2.53~$+$~12\\
7 & 30 & 11 & 1.9063$^{{\rm b}}$ &   7.32~$+$~13 & 8 &  7 &  5 & 151.61 &   6.14~$+$~08 & 8 & 15 &  6 & 1.9171$^{{\rm b}}$ &   3.35~$+$~14\\
7 & 30 & 12 & 1.9080$^{{\rm b}}$ &   1.32~$+$~14 & 8 &  8 &  3 & 106.32 &   5.59~$+$~10 & 8 & 15 &  7 & 1.9193$^{{\rm b}}$ &   1.94~$+$~14\\
7 & 30 & 13 & 1.9115$^{{\rm b}}$ &   2.08~$+$~14 & 8 &  9 &  1 & 78.889 &   1.11~$+$~10 & 8 & 15 &  8 & 1.9210$^{{\rm b}}$ &   6.69~$+$~13\\
7 & 31 &  1 & 1.8395$^{{\rm b}}$ &   9.15~$+$~09 & 8 &  9 &  2 & 83.868 &   1.17~$+$~09 & 8 & 15 &  9 & 1.9298$^{{\rm b}}$ &   6.47~$+$~12\\
7 & 31 &  2 & 1.8442$^{{\rm b}}$ &   1.79~$+$~10 & 8 &  9 &  3 & 84.878 &   8.62~$+$~08 & 8 & 16 &  7 & 1.9115$^{{\rm b}}$ &   1.98~$+$~13\\
7 & 31 &  4 & 1.8483$^{{\rm b}}$ &   5.43~$+$~10 & 8 &  9 &  4 & 91.013 &   1.40~$+$~11 & 8 & 16 &  9 & 1.9219$^{{\rm b}}$ &   5.92~$+$~14\\
7 & 31 &  5 & 1.8505$^{{\rm b}}$ &   1.05~$+$~11 & 8 &  9 &  5 & 106.11 &   1.01~$+$~10 & 9 &  2 &  1 & 974.86 &   1.94~$+$~04$^{\rm g}$\\
7 & 31 & 14 & 1.9081$^{{\rm b}}$ &   4.17~$+$~14 & 8 & 10 &  7 & 86.999 &   1.05~$+$~10 & 9 &  3 &  1 & 93.923 &   7.73~$+$~10$^{\rm g}$\\
7 & 31 & 15 & 1.9120$^{{\rm b}}$ &   2.08~$+$~14 & 8 & 10 &  9 & 115.40 &   1.45~$+$~11 & 9 &  3 &  2 & 103.94 &   2.85~$+$~10$^{\rm g}$\\
8 &  2 &  1 & 1328.9 &   5.00~$-$~01$^{{\rm d}}$ & 8 & 11 &  1 & 1.9176 &   2.73~$+$~14 & 9 &  4 &  1 & 1.9268 &   3.89~$+$~14\\
8 &  3 &  1 & 1118.1 &   1.45~$+$~04$^{{\rm d}}$ & 8 & 11 &  3 & 1.9209 &   1.01~$+$~14 & 9 &  4 &  2 & 1.9306 &   1.94~$+$~14\\
8 &  3 &  2 & 7044.8 &   4.15~$+$~01$^{{\rm d}}$ & 8 & 11 &  4 & 1.9239 &   2.92~$+$~13 &   &    &    &        &            \\
8 &  4 &  1 & 592.23 &   1.74~$+$~04$^{{\rm d}}$ & 8 & 12 &  1 & 1.9154$^{{\rm b}}$ &   2.09~$+$~14 &   &    &    &        &            \\
\hline
\end{tabular}
\smallskip
\begin{list}{}{}
\item[$^{{\rm a}}$]Two-photon transition rate from Drake (\cite{dra86}).
\item[$^{{\rm b}}$]Theoretical wavelength.
\item[$^{{\rm c}}$]Transition rate from Galav\'{\i}s et~al. (\cite{gal98}).
\item[$^{{\rm d}}$]Transition rate from Galav\'{\i}s et~al. (\cite{gal97}).
\item[$^{{\rm e}}$]Transition rate from Mendoza et~al. (\cite{men99}).
\item[$^{{\rm f}}$]Transition rate from Becker et~al. (\cite{bec89}).
\item[$^{{\rm g}}$]Transition rate from C. J. Zeippen, private communication.
\end{list}
\end{table}